Author's response to reviews

Title: Implementing advance care planning: a qualitative study of community nurses' views and experiences

Authors:

Jane Seymour (jane.seymour@nottingham.ac.uk)
Kathryn Almack (k.almack@nottingham.ac.uk)
Sheila Kennedy (sheila.kennedy@nottingham.ac.uk)

Version: 7 Date: 31 March 2010

Author's response to reviews:

In the second quote under the first subheading of the Findings section, a comma is misplaced. It is currently after the word 'here' and needs to be moved to follow the word 'before' i.e.

.....although as people have said here before, it’s something that a lot of district nurses and healthcare professions say.....